\documentclass[12pt]{exam}
\usepackage{amsmath, amsthm, amssymb}
\usepackage{fullpage, paralist}
\usepackage{subfig}
\usepackage{graphicx}
\usepackage{enumerate}
\usepackage{enumitem}
\usepackage{multirow,array}


\newcommand{\Z}{{\mathbb Z}}
\newcommand{\Q}{{\mathbb Q}}
\newcommand{\R}{{\mathbb R}}
\newcommand{\C}{{\mathbb C}}

\newcommand{\A}{{\mathcal A}}
\newcommand{\I}{{\mathcal I}}


\newcommand{\ds}{\displaystyle}
\newcommand{\la}{\langle}
\newcommand{\ra}{\rangle}
\newtheorem{thm}{Theorem}[section]

\theoremstyle{remark}
\newtheorem*{nnremark}{Remark}

\printanswers
%\noprintanswers
\unframedsolutions


\begin{document}

\parskip=\medskipamount \thispagestyle{empty}


\centerline{Math 36 --- Daily Homework}

\centerline{Assigned: 10/11} 

\centerline{Due: 10/13}
  

\begin{questions}

\question 

Write up a description of your final project. 

If you are going to model a real-life situation, you should include the following:

\begin{itemize}
\item Your partner's name, if you have a partner
\item The scenario you are modeling
\item The tools we've learned in class that you will use (i.e. game theory, networks, etc.)
\item The questions you will be trying to answer with your model
\end{itemize}

If you will be studying a particular mathematical technique in more detail, include

\begin{itemize}
\item Your partner's name, if you have a partner
\item The mathematical topic you want to study
\item The names of one or two references and the authors. These could be textbooks about the thing you want to study, or published papers that use the techniques you want to learn.
\end{itemize}


\end{questions}


\end{document}







%%% Local Variables:
%%% mode: latex
%%% TeX-master: t
%%% End: